Author's response to reviews

Title: Suppression subtractive hybridization identified differentially expressed genes in lung adenocarcinoma: ERGIC3 as a novel lung cancer-related gene

Authors:

  Mingsong Wu (hanyue4187@163.com)
  Tao Tu (tutao311@163.com)
  Yunchao Huang (huangych2001@yahoo.com.cn)
  Yi Cao (caoy@mail.kiz.ac.cn)

Version: 4 Date: 4 January 2013

Author's response to reviews: see over
Subject: Manuscript “Suppression subtractive hybridization identified differentially expressed genes in lung adenocarcinoma: ERGIC3 as a novel lung cancer-related gene” (MS: 2134587857791898)

Dear Dr Cho,

I thank you for your letter from January 2, 2013. We appreciate the constructive criticism of reviewers, and have revised the text in line with the suggestion as follows:

Referee: 1

Minor Essential Revisions:

1. The revised manuscript addressed most of my questions and suggestions. The explanation to the nuclear stain signals in immunohistochemistry representative figures is acceptable, but the last sentence of the first paragraph in the Result section on immunohistochemistry shall be written as “this may be the reason that the non-specific staining appeared in nucleus”.

   Accepted and revised. It was coloured in the manuscript.

Referee: 2

No comment.

We thank the reviewers for their comments and hope that our manuscript now will meet the requirements for publication in the BMC Cancer.

Yours sincerely,

Prof. Dr. Yi Cao

Laboratory of Molecular and Experimental Pathology
Kunming Institute of Zoology
Chinese Academy of Sciences
32 Jiaochang Donglu
Kunming, Yunnan 650223
Peoples’ Republic of China
Tel.: 0086-871-5199081
Fax: 0086-871-5199081
E-mail: caoy@mail.kiz.ac.cn